Journal of Business Research xxx (2015) xxx-xxx



Contents lists available at ScienceDirect

### Journal of Business Research



# Entrepreneurial orientation, marketing capabilities and performance: The Moderating role of Competitive Intensity on Latin American International New Ventures\*

Silvia L. Martin<sup>a,\*</sup>, Rajshekhar (Raj) G. Javalgi<sup>b</sup>

<sup>a</sup> California State University, Los Angeles, CA, USA

<sup>b</sup> Monte Ahuja College of Business, Cleveland State University, 2121 Euclid Avenue, Cleveland, OH 44115, USA

#### ARTICLE INFO

Article history: Received 30 September 2014 Received in revised form 31 January 2015 Accepted 25 September 2015 Available online xxxx

Keywords: Competitive intensity International new ventures Entrepreneurial orientation Marketing capabilities Performance

### ABSTRACT

International new ventures (INVs) face constant competitive intensity, which serves to increase the already high difficulty of enhancing marketing capabilities to gain superior performance in foreign markets. International entrepreneurship (IE) literature suggests that entrepreneurial orientation (EO) may influence INVs' performance. However, EO can be a resource-consuming strategic orientation, and INVs face resource constraints. Specifically, the literature expresses doubts as to whether increasing levels of EO are favorable for all INVs, all the time. From the resource-based view (RBV) perspective, the present paper extends previous IE research by investigating whether the degree to which EO and corresponding marketing capabilities vary under differing competitive intensities when enhancing performance. The findings highlight the moderating role of competitive intensity between EO and marketing capabilities for better INV performance. These have important implications for the decisions of IE scholars and practitioners about EO allocation in order to enhance the required marketing capabilities for INV's increased performance.

The survey data consist of Mexican INVs, where the emphasis on constrained resources is high as are the competitive intensity challenges that these firms face in foreign markets. Therefore, this study contributes to the almost wholly ignored research about Latin American new venture internationalization and performance debate.

© 2015 Elsevier Inc. All rights reserved.

### 1. Introduction

Statistics from the International Monetary Fund (IMF) show that emerging markets' share of global gross domestic product (GDP) hit 50.5% in 2013, which correspond to a 31% increase from 1980. Not only have emerging markets increased their share of GDP by an average of 0.6% per year over the past 33 years, but the IMF expects emerging markets' share of global GDP to increase at an even faster pace going forward, to an average of 0.7% per year between now and 2019 to reach 54.5% (IMF 2014). This is a culmination of the increasing emphasis on globalization and the rapid growth of international trade, which have further exhorted firms to seek opportunities for market expansion. The rise of firms from emerging markets that operate internationally accounts for an increasingly important component of the overall market offering (Claessens and Schmukler 2007). For this reason, international entrepreneurship (IE) academics and practitioners have widely acknowledged a nascent paradigm shift of firm internationalization

\* Corresponding author at: College of Business and Economics, California State University, 5151 State University Drive, Los Angeles, CA 90032, USA.

*E-mail addresses*: silvia.martin236@calstatela.edu (S.L. Martin), r.javalgi@csuohiho.edu (R.(R.)G. Javalgi).

studies, from research on developed economies, to investigations based on emerging economies (Cavusgil et al., 2012; Ciravegna et al., 2014).

Over the last decade and a half, emerging market economies, including several countries in Latin America, are regarded as new engines of economic growth (Lederman, et al., 2014). In Latin America, industries and companies have demonstrated that they are at least as innovative and entrepreneurial as their counterparts in the Northern Hemispheres. As Latin America embraces global economic interdependence, smalland medium-sized enterprises continue to play a major role in international trade and investment activities. One way these companies are showing their global orientation is by demonstrating entrepreneurial drive (Frechette, 2006).

Although Latin America is an important world player among emerging economies, the region has been too long neglected among IE literature (Rialp et al., 2005). Four countries of the region (Brazil 7th, Mexico 15th, Argentina 21th, and Venezuela 28th) are in the top 30 economies based on their GDP (World Bank, 2014), and two are in the top four most attractive emerging markets locations (China, Russia, Brazil, and Mexico) (UNCTAD 2014). Nevertheless, less than 4% of IE literature focuses particularly on Latin America (Carneiro and Brenes 2014; Pérez-Batres et al., 2010).

Also, most of the limited research on internationalization and performance of Latin America's firms is centered on established multinationals,

http://dx.doi.org/10.1016/j.jbusres.2015.10.149 0148-2963/© 2015 Elsevier Inc. All rights reserved.

 $<sup>\</sup>star$  The authors are grateful for the comments by anonymous reviewers, Luciano Ciravegna (Royal Holloway, University of London; INCAE), Sumit K. Kundu (Florida International University), and Luis Lopez (INCAE) on prior drafts of this article.

leaving behind new ventures (Carneiro and Brenes 2014; Fleury and Fleury 2014; Salvo 2007). The scarce research about new venture internationalization and performance treats emerging markets as a uniform bloc, without separating Latin America in their conclusions. For instance, Bruton and colleagues' (2008) findings about the accelerating rate of new ventures' internationalization from emerging markets.

In addition, IE scholars have identified some high-tech firms that internationalize rapidly almost from inception known as international new ventures (INVs) (Oviatt and McDougall 1994) or born globals (Knight and Cavusgil 2004). INVs are defined as business organizations that from inception seek to gain significant competitive advantage from the use of resources toward the international sale of outputs (Oviatt and McDougall 1994; Oviatt and McDougall 2005).

Two decades ago, since the seminal articles of INVs, these highly entrepreneurial firms were found often facing constrained resources and high levels of competitive intensity (Oviatt and McDougall 1994). Indeed, INV's behavior over time is distinguished by a strong entrepreneurial orientation (EO). Hence, through a combination of innovativeness, proactiveness and risk-seeking behavior INVs attempt to exploit opportunities by entering the international marketplace quickly (Autio 2005). Nevertheless, recent IE literature (Wiles et al., 2012) suggests that the field is devoid of strong theoretical frameworks related to EO. While the extant literature supports EO as a key ingredient for firm success (Wang 2008), most specifically in INVs (Knight 2000), published research shows skepticism as to whether increasing levels of EO are necessary for significant performance increase (Smart and Conant 1994), or may even lead to poor new venture performance under certain circumstances (Stam and Elfring 2008). Since EO can be a resource-consuming strategic orientation (Wiklund 1999) and INVs are resource constrained (Oviatt and McDougall 2005; Ripollés and Blesa 2012), the degree to determine in which the INV needs EO is essential.

While empirical evidence can be found in support of the impact of environmental factors in performance (Lu and Beamish 2001b), IE literature largely neglects them (Aspelund et al., 2007). Competitive intensity is one environmental factor that has been identified extensively in the fields of strategy (Ramaswamy 2001), innovation (He and Nie 2008), multinational corporations (Cui et al., 2005), exporting (Cadogan et al., 2003), and organizational learning (Auh and Menguc 2005). The competitive intensity that firms face can be defined as the magnitude of effect that a firm has on its rivals' life chances. A weak competitor is one that harms its rivals' life chances only slightly, whereas strong competitors reduce their rivals' life chances dramatically (Barnett 1997). Despite of the significance of competitive intensity, less is known about its effect on the performance antecedents of INVs.

Also, authors have argued recently how the RBV has contributed to the international expansion of INV firms (Ripollés and Blesa 2012; Sapienza et al., 2006). INVs hold a unique constellation of capabilities that enable them to achieve rapid internationalization growth soon after the firm's founding (Knight and Cavusgil 2004; Rialp et al., 2005). Capabilities enable a firm to perform value-creating tasks effectively and they reside in organizational processes and routines that are difficult to replicate (Day, 2011). The literature identifies different capabilities that include technological capability, marketing capability, and operational capability (Krasnikov and Jayachandran 2008). Marketing capabilities can be defined as integrative processes designed to apply the necessary resources the firm possesses to the market-related needs, enabling the firm to add value and meet competitive demands (Day 1994). IE literature shows that marketing capabilities play a major role in INVs (Ripollés and Blesa 2012); however, much work still remains on identifying the best combination of specific marketing capabilities required for the early and accelerated internationalization of young firms to gain superior performance.

Despite the scholarly and managerial attempts to understand the relation between EO and marketing capabilities (Jantunen et al., 2005) to improve INV performance (Martin and Javalgi 2013; Zhou et al.,

2012), the IE literature reveals almost no insights to empirically test these propositions given their complexity. Considering that EO and marketing capabilities are performance antecedents of INVs (Martin and Javalgi 2013) and that competitive intensity is present in different degrees depending on the type of international market, the present study addresses the following still unanswered research question: *What is the contingent effect of competitive intensity in EO, with respect to marketing capabilities and performance of INVs*?

This study makes four contributions to knowledge in this important area of understanding INVs' performance antecedents. First, this study brings together competitive intensity into the RBV-performance framework and set up relationships to develop and empirically test an integrative model of the performance antecedents of INVs. To this end, the findings offer an insightful extension to understanding how the competitive intensity environmental factor comes about and the mechanisms by which INVs can gain from it.

Second, the present investigation studies how the role of EO in conjunction with the RBV can improve performance. Specifically, this paper reports on the interplay of EO when enhancing marketing capabilities and its consequences on the performance of INVs. This research also analyzes the direct path from EO to performance. In doing so, this inquiry demonstrates how the relation of a firm's EO and marketing capabilities realizes superior performance than the simple relation between EO and performance. Thus, the present research offers a solid extension to the IE theory on INVs.

Third, this study focuses on the moderation effect of competitive intensity on two relationships, one between EO and marketing capabilities, and the second between EO and performance on INVs. The novel model will provide needed empirical grounding from which to make recommendations to managers of INVs concerning important resource allocation decisions. Increasing EO levels requires significant resource investments (Wiklund 1999), and managers need to be sure that their investments will gain suitable rewards. Thus far, recommendations to practitioners are uncertain, simply because the performance related consequences of EO on INVs have yet to receive rigorous empirical attention. Given the literature review in this study, doubts are found as to whether increasing levels of EO are favorable for all INVs, under all circumstances. Thus, IE scholars do not know of any conditions under which the benefits of EO outweigh the costs. Similarly, the literature presents lacunas regarding the conditions under which the costs associated with increasing an INV's level of EO outweighs the benefits accrued. This research is timely regarding the possibility that the costs associated with behaving in an entrepreneurial-oriented way in a venture's international operations may be significantly higher than those associated with being entrepreneurial-oriented within a purely domestic setting.

Fourth, this empirical study is administered in the context of hightechnology "born regional" (Lopez et al., 2009) INVs from Mexico. As a Latin American emerging market, Mexico is a unique setting for testing competitive intensity on the performance antecedents of INVs. Hence, the results in this study speak to an important set of firms previously ignored in the competitive intensity performance debate. The present investigation fills in a gap of the under-researched area of literature surrounding INVs from Latin America. Therefore, by examining INVs in Latin America, specifically in Mexico, this inquiry offers a unique setting for the better understanding of INV performance, where the emphasis on constrained resources is higher (Bruton et al., 2008). The act of going international would appear to provide unique challenges that increase the competitive intensity of Mexican INVs with respect to the INVs from foreign markets. Mexico is one of the most open economies in the world, and this has created an adequate arena for high-tech INVs to grow (Wood et al., 2011). The analysis of INVs in an environment of this kind could offer insights beyond previous studies where most of the INVs samples are from developed economies.

Subsequent to this introductory section, the next section outlines the study's theoretical background and states the hypotheses. Then

Section 3 describes the methods chosen to provide evidence on the conceptual framework. Section 4 explains the analysis and reports the results of the hypotheses. Additionally, Section 5 concludes with a discussion of the study's significance and contributions. Finally, Section 6 presents the research limitations.

### 2. Theory and hypotheses

The size and rapid growth of global trade has focused the attention of IE scholars on the internationalization and performance of new ventures from emerging markets (Wood et al., 2011). Although Latin America is an important world player among emerging economies (Carneiro and Brenes 2014), the study of new ventures in the region is almost wholly ignored in the internationalization and performance debate. Evidence show an increasing importance of international trade specially in high velocity industries, where the central role of hightech INVs taking advantage of international trading opportunities has become crucial to understanding the antecedents of performance (Cavusgil and Knight 2015). The present research uses the INV concept to designate these type of firms since INVs are the broadest concept proposed in the literature (Aspelund et al., 2007). As outlined by Coviello and colleagues (2011), Oviatt and McDougall (1994) specifically chose to use the term "international new venture" because many of the firms they observed did not have a global focus. Rather, most competed primarily in their regional markets (Lopez et al., 2009) or in a relatively limited number of countries.

Research connotes that the majority of INV firms begin their overseas activities via exporting (Knight and Cavusgil 2004), particularly when the focus of the firm is on high-technology products (Burgel and Murray 2000). Therefore, consistent with IE literature (e.g., Cavusgil and Zou 1994; Morgan et al., 2012), this study adopts the export venture of the INV firm as the primary unit of analysis. This level of analysis will emphasize the available EO that deploys marketing capabilities for superior performance contingent on competitive intensity.

The IE literature presents contradictions about the beneficial effect of EO for better INV performance (Gabrielsson et al., 2014). On one hand, some literature supports EO as the key driver for strategic initiatives intended to enhance firm performance (Knight 2001). Since firms high in EO are willing to undertake risky decisions, they may more readily accept the uncertainty embedded in further increasing cross-border activity (Lumpkin and Dess 1996). On the other hand, empirical studies do not find significant relationship between EO and performance (e.g., Wiklund and Shepherd 2005).

Some factors to consider in the dispute about the favorable outcome of EO in performance are related with the conceptualization of EO (Slevin and Terjesen 2011) and performance in IE (Hult et al., 2008). While most studies operationalize EO in an export context (Slevin and Terjesen 2011), no consensus is found on how EO should be conceptualized in an international context. Some scholars have modified the wording of the indicators to reflect the internationalization process, and in this manner, they adopt some type of international EO measure (Knight 2001; Kuivalainen et al., 2007). However, researchers employing this route need to be careful not to employ tautological indicators. Slevin and Terjesen (2011) propose that scholars can retain the traditional measure of EO and look at the relationship with international processes. Jones and Coviello (2005) outline this conceptually, and empirical tests include Mostafa and colleagues (2005), as well as Ripollés and Blesa (2007). To advance this dispute, this study employs the traditional measure of EO to explore its effect in performance at the export venture level of INVs.

The mixed results of performance findings derive from the diverse operationalization of the performance construct in IE literature (Ariño 2003). Hult and colleagues (2008) found that more than 90% of studies use only efficient financial performance measures, such as sales-based measures. Nevertheless, they mentioned that as knowledge deepens in the IE field, the use of multiple types of performance measures is

important to gain a more complete view of its nature (e.g., Gomes and Ramaswamy 1999). Thus, besides the traditional effective financial performance, studies should include efficiency and overall operational performance. Thus, this research adopts a multidimensional performance measure to capture efficiency, effectiveness, and adaptiveness at the export venture level of INVs. Efficiency is the outcome of a venture's products and programs in relation to the resources employed in implementing them. The present investigation uses common measures based on profitability such as ROI, ROS, export venture margin, and reaching export venture financial goals in comparison with that of competitors. Effectiveness, is the success of a venture's products and programs in relation to those of its competitors in the market (Vorhies and Morgan, 2003). To measure effectiveness, this study utilizes market share growth, positive changes in the market share, and growth in sales revenue in comparison with that of competitors. Adaptiveness is the venture's success in responding over time to changing conditions and opportunities in the environment (Walker et al., 1987). To capture adaptiveness, this research employs overall export venture performance, number of successful new venture products, time to market for new export venture products, and responding to competitors product changes in comparison with that of competitors. This logic is consistent with the work of Hughes et al. (2010), as they adopted the same performance measure in a positional advantageperformance relationship. The present study also advances from their research by stipulating that the multidimensional performance measure is also applicable in the EO-performance paradigm.

Aspleund and colleagues (2007) present a theoretical comprehensive framework to study performance of INVs. They propose that the factors related to the founding process, such as EO, can lead to performance of INVs. Also, Morgan and colleagues (2012) introduce an extensive international framework of marketing capabilities and export venture performance. Based on the RBV and using an empirical data of exporting firms, they suggest that marketing capabilities play an important role on performance. This research presents progress from Aspleund and colleagues (2007) and from Morgan and colleagues (2012) by stipulating that EO might lead to specific marketing capabilities in the export venture subject to the contingent effect of competitive intensity. EO informs the translation of marketing capabilities to performance in export markets, not without the contingent effect of competitive intensity. Fig. 1 presents the conceptual model. This study argues that EO and the deployment of the required marketing capabilities do not necessarily create enduring high performance, unless competitive intensity moderates the necessary, resource-consuming EO at the export venture level of INVs. While INVs are vulnerable to impediments related to resource limitations to deploy marketing capabilities (Martin and Javalgi 2013), INVs should acquire the essential amount of EO to develop marketing capabilities that the international market is demanding.



Fig. 1. Conceptual framework.

4

# **ARTICLE IN PRESS**

### 2.1. Entrepreneurial orientations of international new venture firms

Cavusgil and Knight (2015) specify that some desirable and promising lines of inquiry to enhance performance of INVs include the understanding of distinct strategic orientations. In particular, one route to superior performance is for INVs to become entrepreneurially oriented. The IE literature holds the proposition that INV's behavior over time is distinguished with a strong EO. This premise was more formally articulated by the two pioneers in this field, McDougall and Oviatt (1994), when they noted that INV's hold a mixture of innovativeness, proactiveness, and risk-seeking behavior across national borders. This brings the INV field closer to the EO of the firm (He and Nie 2008). Indeed, the relevance of EO to the survival and growth of INVs has been widely acknowledged in the IE literature (Kocak and Abimbola 2009; Wiklund and Shepherd 2005).

The conceptualization of EO has been the focus of systematic inquiry in the literature (Covin and Slevin 2006). EO has its roots in the strategymaking process literature (Mintzberg 1973). EO refers to the strategymaking processes that provide organizations with a basis for entrepreneurial decisions and actions (Lumpkin and Dess 1996; Wiklund and Shepherd 2005). Thus, EO can be conceptualized as a firm's strategic orientation and refers to the decision-making activities, processes, and practices that lead to new market entry (Lumpkin and Dess 1996). The salient dimensions of EO can be derived from a review and integration of the strategy and entrepreneurship literatures (Covin and Lumpkin 2011; Miller, 2011). Since Miller's seminal paper (1983), three dimensions of EO have been identified and used consistently in the literature: innovativeness, risk taking, and proactiveness. Innovativeness is the predisposition to engage in creativity and experimentation through the introduction of new products/services as well as technological leadership in new processes. Risk taking involves taking bold actions by venturing into the unknown, borrowing heavily, and/or committing significant resources to ventures in uncertain environments. Proactiveness is an opportunity-seeking, forward-looking perspective characterized by the introduction of new products and services ahead of the competition and acting in anticipation of future demand.

The literature still reveals gaps that ignore Dess and colleagues (1997) call for further integration of EO and the RBV of the firm. The RBV (Barney 1991) has had an influence on the IE literature dialog by helping researchers articulate the drivers of new venture internationalization (Coviello and Cox 2006; Ripollés and Blesa 2012). INVs face non-trivial liability of newness and foreignness in the markets they enter, just as they are mastering internationalization, they must also survive as young firms (Jones and Coviello 2005). Entrepreneurial companies with EO create, define, discover, and exploit opportunities frequently well ahead of their rivals. Given the turbulent environment posed by globalization, INVs necessitate a strong entrepreneurial posture in strategy making. INVs, which may have fewer resources to compete head to head with larger rivals, have in their favor a strong EO and will fare better than those SMEs that lack such an orientation (Cavusgil and Knight 2015).

### 2.2. Entrepreneurial orientation and marketing capabilities

In concert with the RBV line of reasoning, EO as a resource only has potential value. Possession of EO is a necessary but not sufficient condition for value delivery (Barney 1991). A firm needs to take appropriate strategic actions to capitalize on EO in order to gain a competitive advantage and obtain desirable performance (Lisboa et al., 2011). The capabilities by which firms' resources are deployed explain variations in firm performance rather than simple heterogeneity in firms' resources (Eisenhardt and Martin 2000; Morgan et al., 2009b).

Despite the remarkable attention in the literature about the theoretical connections of EO to firm capabilities (Dess et al., 1997), the debate about the empirical implementation of these connections in specific types of firms is considerable (Miller, 2011). Prior studies have tried to incorporate EO into INV literature (Rasmussan et al., 2001). The notion of EO suggests that some firms are more willing than others to continually search for opportunities and solutions outside the realm of their current activities (Lumpkin and Dess 1996; Oviatt and McDougall 2005). Entrepreneurial companies with EO create, define, discover, and exploit opportunities frequently well ahead of their rivals (Chandra et al., 2009). Firms with EO increase their ability to enhance core capabilities (Zahra 2006), and marketing capabilities play a major role in INVs (Ripollés and Blesa 2012). Marketing capabilities can be considered as the skills and competences a firm possesses that help to understand changes taking place in its markets, together with those that enable to operate more effectively in the market place (Day, 1994). Marketing capabilities are the result of an integration process designed to meet the market-related needs of the venture (Vorhies and Morgan 2005).

Empirical evidence from IE literature supports the notion that rapid international involvement triggered by factors related to the founding process, such as EO, requires valuable capabilities to target customers (Aspelund et al., 2007). The rapidly increasing complexity of the international market environment is persuasive. The number of offerings and distribution channels has increased considerably. The forces of market fragmentation and rapid change are everywhere. Marketing capabilities involve speedily developing and launching new products, responding quickly to any customer changes using pricing skills, and providing high levels of support to distributors for delivering customer value (Day, 2011).

In this regard, this study suggests that as young firms venture into foreign markets, they face uncertainty and risks that tend to depict marketing capabilities to enhance performance. Accordingly, this study proposes that the link between EO and marketing capabilities determines how well these capabilities are matched with the INV market requirements in the following hypothesis:

**H1.** Entrepreneurial orientation is positively related to marketing capabilities of the INV performance at the export venture level.

### 2.3. Marketing capabilities and performance

The RBV emphasizes capabilities as central to understanding a firm's performance in foreign markets (Morgan et al., 2006). Additionally, the RBV is based on the assumption of heterogeneity among firms. The more heterogeneous the firms are that compete in the market, the more crucial capabilities become for superior performance (Barney 1991; Makadok 2001; Teece et al., 1997). Marketing capabilities are the processes by which firms select intended value propositions for target customers and deploy resources to deliver these value offerings in pursuit of desired goals (Day, 2011; Vorhies and Morgan 2005). The literature suggests that marketing capabilities in particular may be immobile (Capron and Hulland 1999), inimitable, and largely non-substitutable value-creation mechanisms (Morgan et al., 2009b).

The firm is able to use marketing capabilities to be better positioned to rapidly and successfully launch and deliver new products, use pricing skills to respond quickly to any customer changes, deliver high-quality after-sales service, and work closely with distributors and retailers in the market (Day, 2011).

Firms expend significant resources on building, maintaining, and leveraging marketing capabilities, and recent research has suggested the link of marketing capabilities and firm performance (Krasnikov and Jayachandran 2008; Vorhies and Morgan 2005). Theoretically, such interdependency may make marketing capabilities a more inimitable resource and therefore a greater potential source of competitive advantage (Barney 1991). Theory assumes that managers not only can isolate distinct marketing capabilities they believe to be valuable, but also can empirically link these capabilities with superior performance (Morgan et al., 2009a).

Zhou and colleagues (2012) identified marketing capabilities in determining the international growth of early internationalization. Prior exporting research has suggested that different marketing capabilities may be most valuable to firms in combination as they interact in ways that help firms achieve superior performance (Ramaswami et al., 2009). Also, Morgan and colleagues (2012) empirically investigate in the export marketing context and on the basis of the RBV that firms operating in such markets are typically more heterogeneous than firms in domestic markets. As a result, marketing capabilities are crucial for leveraging positional advantages and export performance (Zou et al., 2003). Thus, firms can improve their performance by focusing on such marketing capabilities (Weerawardena et al., 2007). In response, and in light of this discussion, this research proposes:

**H2.** The possession of marketing capabilities is positively related to INV performance at the export venture level.

### 2.4. Entrepreneurial orientation and performance

Despite entrepreneurial firms seeking to identify and seize opportunities in the marketplace (Slater and Narver 1994), contemporary IE literature (Wiles et al., 2012) advises that the field is lacking of strong theoretical frameworks related to EO. In particular, Keupp and Gassmann (2011) argue that research in the field is largely phenomenological and studies capturing the EO of the firm are underrepresented. Consequently, this is a contradiction in the INV field; on the one hand, the EO of these firms is taken for granted, while on the other hand, studies that examine EO in INVs are lacking.

International markets are environments of rapid change, shortened product, and business model lifecycles, where the future profit streams from existing operations are uncertain and businesses need to constantly seek out new opportunities. Therefore, firms may benefit from adopting an EO because efforts to anticipate demand to position new product/service offerings often result in strong performance (Ireland et al., 2003). The EO-performance literature is extensive, and with some tendency to regard that firms with more EO have superior performance (Wiklund and Shepherd 2005; Zahra and Covin 1995), although the empirical findings are not altogether consistent. Thus, conceptual arguments suggest that EO leads to higher performance. However, the magnitude of the relationship seems to vary across studies. While some studies have found that businesses that adopt a strong EO perform much better than firms that do not adopt an EO (Hult et al., 2003; Lee et al., 2001; Wiklund and Shepherd, 2003), other studies reported lower correlations between EO and performance (Dimitratos et al., 2004; Lumpkin and Dess 2001; Zahra 1991), or were even unable to find a significant relationship between EO and performance (George et al., 2001; Wiklund and Shepherd 2005). Thus, the literature indicates a considerable variation in the reported relationships between EO and business performance. These contradictory results about the effects of EO on performance have a higher impact on resource-constrained INVs because they need to overcome their limitations to commit an adequate combination of resources and capabilities in their early internationalization.

This study aims to fill this gap by providing empirical evidence drawn from INVs. In line with earlier suggestions, this research considers EO with manifestations of innovative, proactive, and risk-taking behaviors abroad. Since firms high in EO are willing to undertake risky decisions, they may more readily accept the uncertainty embedded in further increasing cross-border activity. Furthermore, the perceived uncertainty in foreign markets may be followed by the constant seeking of new opportunities (Shrader et al., 2000). Therefore, when a firm proactively reflects the inertia for exploiting emerging opportunities, experimenting with change and mobilizing first-mover actions, a firm may overcome the uncertainty that arises when increasing the intensity of its activities in that market (Jones et al., 2011). Similarly, the degree of

innovative activity may also increase the firm's potential to leverage its existing capabilities by increasing the intensity of its activities in current foreign markets or by entering new foreign markets (Cavusgil and Knight 2015).

Rauch and colleagues (2004), based on a meta-analysis of 37 studies, conclude that the EO-performance relationship is moderately large and that firms benefit from EO. From a conceptual standpoint, research variously describes EO as an antecedent to various international phenomena, including exporting (Dess et al., 1997), internationalization (Knight 2001), and export performance (Dess et al., 1997). Based on the previous discussion that supports EO for opportunity recognition in new markets, the literature suggests a reason to suppose that EO has a positive effect on international performance. With reference to the previous, this research suggests.

**H3.** Entrepreneurial orientation is positively related to INV performance at the export venture level.

#### 2.5. The moderation role of competitive intensity

Competitive intensity is one of the factors contributing to environmental hostility (Zahra and Covin 1995). Competitive intensity is a situation where competition is fierce due to the number of competitors in the market and the lack of potential opportunities for further growth. As competition further intensifies, the results of a firm's behavior will no longer be deterministic but random as the behavior is heavily influenced by the actions and contingencies undertaken by competitors. Thus, under conditions of intensifying competition predictability and certainty diminishes (Auh and Menguc 2005).

The possibility of a moderating effect of competitive intensity is consistent with a long tradition of support for the theory that environment moderates the effectiveness of organizational characteristics (Slater and Narver 1994). For example, numerous studies (Auh and Menguc 2005; Cadogan et al., 2003; Cui et al., 2005; He and Nie 2008; Ramaswamy 2001) have found that the effectiveness of a particular strategic orientation is contingent with the dynamics of the market. Scholars have argued that aspects of the environment may moderate the firms' performance antecedents (Cadogan et al., 2003; Jaworski and Kohli 1993).

When the competition is less intense, firms can operate with their existing systems to fully capitalize on the transparent predictability of their own behavior. However, when competition is intense, firms will need to engage in risk-taking and proactive activities to adapt accord-ingly (Cui et al., 2005). To this end, Zahra (1993:324) stated, "When rivalry is fierce, companies must innovate in both products and processes, explore new markets, find novel ways to compete, and examine how they will differentiate themselves from competitors."

This study extends this framework into the INV firms and asserts that this contingency effect will differ between EO and marketing capabilities. This research proposes that with high competitive intensity, the INV will need to engage in risk-taking and proactive activities, to rapidly develop and launch successful new export venture products. These can be achieved by using pricing skills to respond quickly to any customer changes and also to deliver high-quality after-sales service by attracting a retaining after-sales service personnel. In addition, it attracts and retains the best distributors in the export venture market. When the INV engages in risk-taking and proactive activities, it means more EO to enhance more marketing capabilities such as new product development, sales, and distribution to differentiate themselves from competitors. When competition is not intense, INVs will not need so much EO to intensify marketing capabilities for superior performance because of the more predictability in the markets. Based on the earlier discussion, this study proposes the following hypothesis:

**H4.** The relationship between EO and marketing capabilities is moderated by competitive intensity in the INVs' environments. Specifically,

the relationship diminishes under conditions of low competitive intensity. As competitive intensity grows, the relationship becomes stronger.

An EO has been found to be positively associated with performance (Hult et al., 2003; Lee et al., 2001; Wiklund and Shepherd 2003), although empirical findings are not altogether consistent. Some studies reported weaker links (Dimitratos et al., 2004; Lumpkin and Dess 2001; Zahra 1991) and still others did not find a significant relationship between the two constructs (Covin et al., 1994; George et al., 2001; Wiklund and Shepherd 2005). The mixed empirical evidence limits the understanding of the EO–performance relationship.

Lumpkin and Dess (1996) considered the relationship with performance to be context-specific. Additionally, the findings of Zahra and Gravis (2000) highlight the importance of entrepreneurial activities for success in general, but also on international markets as an entrepreneurial orientation supports opportunity recognition. They suggest that entrepreneurially oriented firms actively seek new operating modes and methods that improve performance. Firms need to engage in a greater level of entrepreneurial activities, such as innovation, exploration, and strategy renewal as environmental hostility intensifies (Zahra and Covin 1995). In the present study, this framework is extended into the INV firms and assert that this contingency effect will differ between EO and performance. Based on the previous reasons, this research proposes that with high competitive intensity, the INV will be required to engage in innovation, risk taking, and proactive activities to maintain or enhance performance. With less competitive intensity, the INV firm will not need to incur in such resource-consuming activities. Based on the earlier discussion, this research suggests the following hypothesis:

**H5.** The relationship between EO and performance is moderated by competitive intensity in the INVs' environments. Specifically, the relationship diminishes under conditions of low competitive intensity. As competitive intensity grows, the relationship becomes stronger.

### 3. Research methodology

Following the increased attention in Latin America to the role of INVs (Lopez et al., 2009), data were collected in Mexico for this study. As one of the most open economies in the world, Mexico has created an adequate arena for high-tech INVs to grow (Wood et al., 2011). Mexico is an export-oriented economy, which is the result of significant public policy efforts to generate an open economic process and to diversify the export products looking to raise the participation of industry sectors other than petroleum, identifying high technology as a vital sector (Gray and Cuevas 2005). Mexico has negotiated more than 12 free trade agreements with 44 countries from North America, Europe, Latin America, and Asia. This net of agreements offers preferential access to a superior market of 1.3 billion of consumers. According to Goldman Sachs, Mexico is also included in the BRIMC classification, a relatively new marketing term to refer Brazil, Russia, India, Mexico, and China. The BRIMC acronym derives from the investment bank thesis called BRIC, referring to the fast growing developing economies (O'Neill et al., 2005).

The sampling frame of this study consisted of 260 INVs from Mexico, from a total population of 1433 INVs found. High-technology exports from Mexico have grown substantially over the past decade following an extensive program of trade liberalization (Aulakh et al., 2000). However, according to the current literature review, INVs in Mexico had not been identified as such before the present study. INVs were categorized just as exporting firms without a high-tech identifier in the government databases of exporting firms from Bancomext (Banco Nacional de Comercio Exterior) and Nafinsa (Nacional Financiera). These firms were mixed with other non-high-tech industries in such databases. Thus, the requirement to develop this research was to generate firm –and international venture– level data from high-technology industries in Mexico. In doing so, the databases were selected and

evaluated in terms of the information content across high-technology industries (Fernhaber et al., 2007). The databases were assessed in terms of the inclusiveness of fields enabling the identification of hightechnology INVs in Mexico, such as the age of the firm at exporting, the industry sector, and the firm size in terms of number of employees and sales turnover. A multi-industry sample for this study was used to strengthen the generalizability of the findings and to increase observed variance (Autio et al., 2000; Knight and Cavusgil 2004; Moen 2002; Zahra et al., 2000). Oviatt and McDougall (1994: 49) define an INV firm as "a business organization that from inception seeks to derive significant competitive advantage from the use of resources and the sale of outputs in multiple countries." Although this definition implies that an INV is international at inception, most scholars do not interpret this literally. Instead, they typically view the definition as more descriptive and examine firms that internationalize within their first few years of existence. For example, some studies examined the internationalization of new venture firms that were up to 6 years old (Shrader 2001; Zahra et al., 2000), as being similar to domestic new ventures (Fernhaber et al., 2007). Other studies identified samples of INVs that internationalized within 3 years of their founding (Knight and Cavusgil 2004), and still others used firms that began exporting within 2 years of their founding (Moen 2002). The latter definition of 2 years was adopted. Additionally, to identify Mexican high-technology firms, the classification of the American Electronics Association was followed.

The European Commission classification of SMEs regarding firm size (which is in line with the Mexican Ministry of Economy) was adopted. Firms with 10–50 employees are considered small, and firms in the range of 51–249 employees are medium sized. Firms with fewer than 10 employees are micro firms and were omitted from the study; such firms tend to have part-time operations and unstable objectives that can skew study outcomes.

After drawing together multiple databases and making one of just INVs, a target population of 1422 Mexican INVs was grouped together for this study. Then each firm was checked as part of a screening process to determine the usable sampling frame. After screening for trading status, contact details, firm characteristics, and willingness to participate, largely because of company policies of noncooperation, 111 firms were dropped. This screening process led to a usable sample frame of 1311 INVs. Also, each firm was pre-notified and the identity of the key informant was obtained. The 1311 firms' representatives had expressed willingness to participate. A process was then begun of computer-assisted random calling of the firms in the usable sample frame to conduct telephone interviews to obtain responses for the survey.

The Ethic code for Market Research and Public Opinion in Mexico (AMAI 2008) was followed. After resource exhaustion and the elapse of time, a total of 260 INVs had participated in the study. Each respondent reported on a self-identified export venture, which this study defined as a single product or product line exported to a specific export market (country). The final response rate was 19.8% of the usable sample population. The specific export markets of the Mexican INVs' sample was 70% North America, 21% Central and South America, 4% Europe, 3% Asia, 1% East Europe, and 1% Middle East and Africa. Following Hair and colleagues (2006), 260 observations are considered adequate for analysis purposes. Most of the respondents identified themselves as executive managers or managers (78%); the remainder were executive directors (11%), chief executive officers (5%), or in other senior positions (6%). The mean relevant working experience of the respondents was 6.8 years. A post hoc competency check on the informants' knowledge of export venture marketing programs, competitive strategies, and marketing capabilities, as well as those of their major competitors, elicited a mean of 6.00 on a seven-point scale (1 = "low knowledge," and 7 = "high knowledge"). The export ventures ranged across the following high-technology sectors: 25% computer systems design and related services, 13% computer and peripheral equipment, 18% electronic components, 22% communications equipment, and 22% measuring and control instruments. The median

number of years of exporting was 13. Regarding size, 174 firms had 51 employees or more, and the remaining 86 had 10–50 employees. To assess potential non-response bias, early and late respondents were compared with respect to various characteristics, including number of full-time employees, years of exporting, annual sales volume, age of the venture, number of export markets, key informant self-reported competency evaluation indicators, and the construct measures. No significant difference was detected using secondary information on employee numbers and annual sales volume. The respondent firms were also compared to a group of 70 randomly selected nonparticipant firms. No differences were found between respondents and non-respondents at conventional levels (p < .05). The conclusion was that non-response bias was not a significant problem in the data. A systematic questionnaire development process was used combining fieldwork and literature-based insights to specify the domain of each of the

constructs and to develop multiple items to serve as indicators. An extensive literature search was performed to develop a preliminary survey instrument in English; then five academic researchers in international business serving as expert judges evaluated the survey to assess face validity. Subsequently, a Spanish version of the questionnaire was developed, including business context terms used in Mexico. Two language experts were used to perform a back-translation procedure. Finally, to evaluate the relevance of the constructs to the Mexican INV business environment and the clarity of instructions and response format, the survey was presented and revised in a series of face-to-face settings with nine Mexican INV managers. All construct measures were retrieved from literature existing sources. EO, marketing capabilities, and performance are second-order reflective constructs with three or more dimensions each. Competitive intensity was treated as a first-order construct following Morgan and colleagues (2004) suggestion.

#### Table 1

Measurement models and measures

Factors and items	Standardized loading	<i>t</i> -value
Measurement Model 1: marketing capabilities		
New product development $(CR = 0.93, AVE = 0.82)^a$	0.59 <sup>b</sup>	
Developing new export venture <sup>a</sup>	0.74 <sup>b</sup>	
Successfully launching new export venture products	0.87	13.05
Speedily developing and launching new export venture products	0.86	12.97
<b>Sales</b> $(CR = 0.89, AVE = 0.64)^a$	0.82	4.7
Using our pricing skills to respond quickly to any customer changes	0.51 <sup>b</sup>	
Delivering high-quality after-sales service	0.75	7.16
Attracting and retaining after-sales service personnel	0.80	7.21
<b>Distribution</b> $(CR = 0.82, AVE = 0.73)^{a}$	0.74	5.4
Providing high levels of support to distributors	0.72 <sup>b</sup>	
Closeness in working with distributors/retailers in this market	0.63	8.99
Adding value to distributors' businesses	0.70	9.92
Satisfying the needs of distributors	0.70	9.94
Attracting and retaining the best distributors in the export venture market	0.68	9.65
Goodness-of-fit Statistics: $\chi^2/df = 82.783$ (42), $p < 0.01$ ; GFI = 0.95; NFI = 0.93; CFI = 0.96; TLI = 0.96; T	.95; RMSEA = 0.063.	
Measurement Model 2: performance construct and competitive intensity construct		
Performance construct		
Efficiency ( $CR = .94$ , $AVE = .80$ ) <sup>c</sup>	0.86 <sup>b</sup>	
Return on investment (ROI)	0.79 <sup>b</sup>	
Return on sales (ROS)	0.76	12.51
Export venture margin	0.80	13.31
Reaching export venture financial goals	0.73	12.01
Effectiveness $(CR = .96, AVE = .90)^{\circ}$	0.69	7.59
Positive changes in market share	0.79 <sup>b</sup>	
Market share growth	0.87	14.57
Growth in sales revenue	0.84	14.11
Adaptiveness ( $CR = .88$ , $AVE = .66$ ) <sup>c</sup>	0.87	7.08
Overall export venture performance	0.64 <sup>b</sup>	
Number of successful new export venture products	0.73	9.04
Time to market for new export venture products	0.76	9.26
Responding to competitors product changes	0.48	6.56
<b>Competitive intensity construct</b> $(CR = .87, AVE = .70)^{c}$		
Competition is cutthroat	0.93 <sup>b</sup>	
There are many promotion wars	0.72	9.46
Aggressive selling is the norm	0.62	8 70
Goodness-of-fit statistics: $\chi^2(df) = 127.896$ (73), $p < 0.01$ ; GFI = 0.94; NFI = 0.92; CFI = 0.97; TLI =	= 0.96; RMSEA $= 0.054$	
Measurement Model 3: entrepreneurial orientation construct		
Innovativeness (CR = $0.89$ , AVE = $0.76$ ) <sup>c</sup>	0.67 <sup>b</sup>	
Technical innovation based on research results is readily accepted	0.78 <sup>b</sup>	
Management actively seeks innovative ideas	0.97	11.31
Innovation is readily accepted in program/project management	0.35	5.52
<b>Riskiness</b> (CR = $0.81$ , AVE = $0.60$ ) <sup>c</sup>	0.97	5.67
Conservative with major decision	$0.66^{\mathrm{b}}$	
New projects are approved stage by stage	0.48	5.66
Support projects where expected returns are certain	0.48	5.63
<b>Proactiveness</b> $(CR = 0.70, AVE = 0.55)^{c}$	0.5	
Look out for business	0.97 <sup>b</sup>	
First to introduce new brands	0.4	2.82
Goodness-of-fit statistics: $\chi^2$ (df) = 26.495 (17), $p < 0.01$ ; GFI = .98, NFI = 0.95; CFI = 0.98; RMSEA	=0.046	

Note: CR = composite reliability, AVE = average variance extracted.

<sup>a</sup> Anchored by 1 = "much worse" and 7 = "much better."

<sup>b</sup> Fixed to set the scales.

 $^{\rm c}~$  Anchored by 1 = "not at all" and 7 = "to a great extent."

The concept of EO reflects the firm-level processes, practices, and decision-making style (Lumpkin and Dess 1996) of entrepreneurial organizations. In its commonly used form, EO is an aggregate measure consisting of three dimensions: innovativeness, proactiveness, and risk taking (Covin and Slevin 1989; Wiklund 1999). Innovativeness was measured with items from Hurley and Hult (1998), additionally items from Morgan and Strong (2003) were used to measure riskiness and proactiveness. A Likert-type seven-point scale was employed ranging from (1) "not at all" to (7) "to a great extent" with a mid-point label of "To some extent."

The dimensions of marketing capabilities are new product development, sales, and distribution. While items from Zou, Fang, and Zhao, (2003) were used to capture new product and distribution capabilities, Katsikeas, Paparoidamis, and Katsikea (2004) items were used to source sales capabilities. Likert-type seven-point scale was employed to operationalize marketing capabilities ranging from (1) "much worse" to (7) "much better" with a mid-point label of "about the same."

Multidimensional measures of performance should be employed in the field of IE (Hult et al., 2008; Robson et al., 2008). Accordingly, INV performance is conceptualized in this study at the export venture level in terms of three dimensions used by Jaworski and Kohli (1993) and Walker and colleagues (1987): (1) efficiency, the relationship between performance financial outcomes and the inputs required to achieve them; (2) effectiveness, the extent to which organizational goals and objectives are met; and (3) adaptiveness, the operational ability to respond to environmental changes. Effectiveness and efficiency items were obtained from Vorhies and Morgan (2003) and Walker and Ruekert (1987). Walker and Ruekert (1987) were used to measure adaptiveness items. The participating INV managers were asked to provide their own rating of their firm's performance relative to the major competitors using a Likert-type seven-point scale, ranging from (1) "much worse" to (7) "much better" with a mid-point label of "about the same." As well, Cadogan and colleagues (2003) adaptations of Jaworski and Kohli (1993) measures of competitive intensity were used to reflect changes taking place in export markets and not changes taking place in domestic markets.

### 4. Analysis and results

Three measurement models were estimated (see Table 1). The measurement model analyses were performed using the elliptical reweighted least squares estimation procedure in AMOS, which is proved to produce unbiased parameter estimates for multivariate normal and non-normal data. Despite a significant chi-square ( $\chi^2 = 82.78$ ; df = 42; p < .01) in the first measurement model, as might be expected given the sensitivity of the test statistic to sample size (Bagozzi and Yi 1988), all other diagnostics are supportive. The chi-square ratio ( $\chi^2/df = 1.97$ ) is aligned with the adequate fit for minimum discrepancy of Byrne (1989). The other fit indexs (goodness-of-fit index [GFI] = .95, normed fit index [NFI] = .93, comparative fit index [CFI] = .96, Tucker-Lewis index [TLI] = .95, and root mean square error of approximation [RMSEA] = .063) suggest that the model fits the data

Table	2	
_		

Descriptives and correlations.

satisfactorily. Items loaded heavily on their posited constructs with values greater than .5 (t-values >4.6).

Likewise, the second measurement model exhibits a good overall fit to the data (GFI = .94, NFI = .92, CFI = .97, TLI = .96, RMSEA = .054) despite a significant chi-square ( $\chi^2 = 127.89$ ; df = 73; p < .01) (Bagozzi and Yi 1988), as might be expected given the sensitivity of the test statistic to sample size. The chi-square ratio ( $\chi^2/df = 1.75$ ) is aligned with the adequate fit for minimum discrepancy of Byrne (1989). Items loaded heavily on their posited constructs with value greater than .47 (*t*-values >6.4). The third measurement model shows good fit values (GFI = .98, NFI = .95, CFI = .98, TLI = .97, RMSEA = .046) with a chi-square ratio ( $\chi^2/df = 1.56$ ) aligned with the adequate fit for minimum discrepancy of Byrne (1989). Unidimensionality is also obtained in all measurement models based on the good fit values of the fit statistic.

The measurement models themselves offer support for convergent validity if the overall goodness-of-fit indexes demonstrate a good fit of the hypothesized relationships to the data and all factor and item loadings are high and significant (Anderson and Gerbing 1988). In general, the results exhibit a good fit of the measurement models to the data and high standardized loadings significant at p < .01. Furthermore, average variance extracted (AVE) estimates for the measures range from .55 to .90 (see Table 1). Composite reliability coefficients for all scales range from .70 to .96, suggesting satisfactory internal consistency. Fornell and Larcker's (1981) test of discriminant validity was employed. This procedure involves assessing whether the AVE for every construct's measure is larger than the squared phi correlation of that construct with all other constructs in the model. All AVE estimates compare favorably with the corresponding squared phi correlations. Table 2 presents the Pearson's correlations and descriptive statistics of the measures. In summary, the measures possess adequate psychometric properties.

### 4.1. Structural model and results

To test the hypotheses, the parsimonious structural model estimation procedure was used for this study. The parsimonious approach entails averaging the indicators for each construct to form manifest composites. By conducting such a procedure, the first-order construct is represented by one single indicator, and the second-order constructs are treated in the model as being first-order with composites of their dimensions (Morgan et al., 2004). Second-order constructs are presented in the model as composites of their dimensions. Because the number of parameters estimated relative to sample size is a key determinant of convergence, standard errors, and model fit, this method was critical in achieving a ratio of sample size to estimated parameter greater than five, which is necessary to attain reliable parameter estimates (Bentler 1995). As such, composite measures were used as manifest indicators for each latent construct by averaging the items of each subscale. Additionally, checking visually if the additional level satisfies the t-rule of identification, e.g., the number of data variances and covariances equals or exceeds the number of parameters to be estimated (Byrne 2001), is crucial to modeling higher-order constructs. Each

Measure	М	SD	1	2	3	4	5	6	7	8	9
1. New Product	4.85	1.61	0.91								
2. Sales Service	5.16	1.31	.441	0.80							
3. Distribution Service	4.77	1.33	.357	.477	0.86						
4. Efficiency	4.68	1.50	.274	.364	.372	0.90					
5.Effectiveness	5.43	1.23	.222	.403	.275	.515	0.95				
6. Adaptiveness	4.77	1.44	.438	.458	.328	.589	.483	0.81			
7. Innovativeness	5.00	1.28	.173	.319	.172	.309	.258	.287	0.87		
8. Riskiness	5.24	1.16	.161	.275	.119	.312	.225	.432	.395	0.77	
9.Proactiveness	5.13	1.31	.265	.241	.085	.250	.382	.356	.196	.294	0.74

Note: Correlations  $\geq$ .12 or  $\leq$ .12 are significant at  $p \leq$ .05 (two-tailed). Bold diagonal elements are the square root of AVE.

#### S.L. Martin, R.(R.)G. Javalgi / Journal of Business Research xxx (2015) xxx-xxx

### Table 3

Structural model.

Structural relationships	Standardized loading	<i>t</i> -value
Hypothesized links		
<b>H1</b> Entrepreneurial orientation $\rightarrow$ marketing capabilities	.51	4.34**
H2 Marketing capabilities → performance	.56	4.64**
H3 Entrepreneurial orientation → performance	.48	4.37**
H4 Split group moderation test		
Low competitive intensity group		
Entrepreneurial orientation $\rightarrow$ marketing capabilities	.30	2.31*
High competitive intensity group Entrepreneurial orientation → marketing capabilities H5 Split group moderation test Low competitive intensity group Entrepreneurial orientation → performance	.91	4.14 <sup>**</sup> 3.21
High competitive intensity group Entrepreneurial orientation $\rightarrow$ performance Goodness-of-fit statistics: $\chi^2(df) = 80.643(31)$ , p < .01; NFI = .88; CFI = .92; RMSEA = 0.079	.47	1.11

Note: critical value ( $\alpha = .5$ ) = 1.645.

\*  $p \leq .05$  (one-tailed, directionality hypothesized).

\*\*  $p \leq .01$  (one-tailed, directionality hypothesized).

construct and any structure was checked if an additional constraint was required.

The control variable in the structural model is firm size, measured in terms of number of employees. Hence, a path from firm size to performance was included. Given the relatively large sample, the significant chi-square is not surprising ( $\chi^2 = 80.64$ ; df = 31; p < .01). The fit indexes (GFI = .94, NFI = .88; CFI = .92; RMSEA = .079) suggest the structural model demonstrates a good fit to the data (see Table 3).

The empirical assessment of key relationships predicted in the theoretical model (Fig. 1) indicates support for all relationships examined with exception of one (Table 3). The results show that EO presents a positive significant relation with marketing capabilities supporting H<sub>1</sub> ( $\beta$  = .51, p < .01). The relation between marketing capabilities and performance displays a significant loading path giving strong support to H<sub>2</sub> ( $\beta$  = .56, p < .01). In support of H<sub>3</sub>, the results uphold that EO is positively linked to performance ( $\beta$  = .48, p < .01).

An additional analysis was required to test that competitive intensity moderates the relationship between EO and marketing capabilities. The sample was split into two groups at the median level of competitive intensity and the structural model was re-estimated (Hewett and Bearden 2001). Two models were estimated: one in which the path between EO and marketing capabilities was constrained to be equal across the two groups and one in which the path coefficients were allowed to vary freely. A highly significant chi-square difference ( $\Delta \chi^2_{(1)} = 13.02$ , p < .01) signifies much better fit for the unconstrained model, thus indicating that the relationship between EO and marketing capabilities is different in the two groups. As shown in Table 3, the two-group moderator test supports the prediction of the theoretical model H<sub>4</sub>. In the low competitive intensity group, the EO and marketing capabilities relationship is positive and significant (path coefficient = .30, *t*-value = 2.31, p < .05) and in the high competitive intensity group, the same relationship is positive and significant (path coefficient = .91, *t*-value = 4.14, *p* < .01).

To test that competitive intensity moderates, the relationship between EO and performance another set of models was estimated: one in which the path between EO and performance was constrained to be equal across the two groups and one in which the path coefficients were allowed to vary freely. The results exhibit a non-significant chi-square difference in the two groups  $(\Delta \chi^2_{(2)} = 1.04, p > .05)$ . Therefore, the path from EO and performance is not moderated by competitive intensity failing to support H<sub>5</sub> (see Table 3).

#### 5. Discussion and conclusions

Since the seminal work of Oviatt and McDougall (1994), the study of INVs has become increasingly significant in the IE arena. This study is an attempt to address the lacuna in IE theory on the interplay among EO, marketing capabilities, competitive intensity, and performance of INVs. The study findings support all the hypotheses with exception of  $H_5$  and signify the efficacy of the measurement approaches used to capture the focal constructs. Yet, until now, no study has examined weather competitive intensity moderates the relationship between EO and marketing capabilities on the export venture performance of high-technology INVs. This study makes four valuable contributions to knowledge as a result.

First, the present research contributes to the RBV-performance framework by bringing forward competitive intensity to extend and empirically prove new relationships in an integrative model to comprehend the performance antecedents of INVs. Recent investigations observe that limited focus is given to environmental factors, such as competitive intensity, in the theoretical models of INVs' performance (Aspelund et al., 2007). Thus, this study's findings offer a novel extension to the understanding of how competitive intensity comes about and how high-technology INVs can gain from them. Second, the findings highlight the interplay between EO and marketing capabilities to demonstrate how performance is realized. The results show that the path from EO to marketing capabilities enhances higher performance, than the direct path from EO to performance.

Third, competitive intensity moderates the relationship between EO and marketing capabilities of INVs. If the competitive intensity is higher, EO becomes a key component for INVs to enhance marketing capabilities. In contrast, EO and marketing capabilities may be not as relevant under conditions of low competitive intensity. Under higher competitive intensity conditions, INVs with less EO and marketing capabilities are likely to see their performance impaired as customers switch to firms with more marketing capabilities. Thus, the possibility of a high level of EO and marketing capabilities is not always desirable given that its development and maintenance are highly resource intensive (Slater and Narver 1994) and that the rewards from having a high level of EO and marketing capabilities may not always accrue. While competitive intensity moderates between EO and marketing capabilities, it does not moderate in the path of EO to performance. Thus, the degree of competitive intensity could determine the required EO to leverage marketing capabilities for superior performance. However, competitive intensity is not useful to determine the amount of EO required for superior performance when marketing capabilities are not present as a mediator of the EO-performance relationship.

Fourth, the results speak to an important set of firms hitherto ignored in the debate of the EO–performance paradigm and marketing capabilities: the high-technology INV. This study fills in a gap of an under-researched area of INVs from Latin America. Thus, in the context of INVs analyzed at the export venture level, EO and marketing capabilities with competitive intensity as a moderator have synergistic value-creating effect over performance. Therefore, EO and marketing capabilities are important causal mechanisms moderated by competitive intensity that help explain performance in INVs.

### 6. Limitations of the study

The first limitation of this study is the cross-sectional research design, which prohibits causal inference, and temporal effects exist among EO, marketing capabilities, performance, and competitive intensity that are not accommodated within this empirical framework. Further research should be aimed at generating longitudinal data to capture dynamic influences. However, this limitation is common in studies conducted within the area of accelerated internationalization (Freeman and Cavusgil 2007). Second and partly related to the latter, reverse causation cannot be ruled out in the theoretical framework of this

## <u>ARTICLE IN PRESS</u>

S.L. Martin, R.(R.)G. Javalgi / Journal of Business Research xxx (2015) xxx-xxx

study. Third, the unit of analysis in the study was the export venture of the INV firm, identified by the respondent. No accommodations were made to assess the related or discrete effects of EO, marketing capabilities, performance, and competitive intensity with regard to other concurrent and historical ventures. Fourth, these data were generated among the INVs of a single country: Mexico, that exports to different countries. Therefore, the results are limited to this particular country's framework and caution should be exercised in attempting to draw generalizations to other contexts. Fifth, a multi-industry sample was used to increase generalizability, but in doing so, the sample becomes heterogeneous, and the ability to represent a single industry closely was lost. Nevertheless, these multiple industries are all high-technology oriented. Collectively, then the findings are limited by these features of the sample.

### References

- AMAI (2008). Código de ética para la investigación de mercados y opinión pública en México [ethics code for market research and public opinion in Mexico]. Mexico City: Asociación Mexicana de Agencias de Investigación, AMAI.
- Anderson, J.C., & Gerbing, D.W. (1988). Structural equation modeling in practice: A review and recommended two-step approach. *Psychological Bulletin*, 103(March), 411–423.
- Ariño, A. (2003). Measures of strategic alliance performance: An analysis of construct validity. Journal of International Business Studies, 34(1), 66–79.
- Aspelund, A., Madsen, T.K., & Moen, O. (2007). A review of the foundation, international marketing strategies, and performance of international new ventures. *European Journal of Marketing*, 41(11/12), 1423–1448.
- Auh, S., & Menguc, B. (2005). Balancing exploration and exploitation: The moderating role of competitive intensity. *Journal of Business Research*, 58(12), 1652–1661.
- Aulakh, P.S., Kotabe, M., & Teengen, H. (2000). Export strategies and performance of firms from emerging economies: Evidence from Brazil, Chile and Mexico. Academy of Management Journal, 43(3), 342–361.
- Autio, E. (2005). Creative tension: The significance of Ben Oviatt's and Patricia McDougall's article 'toward a theory of international new ventures'. *Journal of International Business Studies*, 36(1), 9–19.
- Autio, E., Sapienza, H.J., & Almeida, J.G. (2000). Effects of the age at entry, knowledgeintensity, and imitability on international growth. Academy of Management Journal, 43(5), 909–924.
- Bagozzi, R.P., & Yi, Y. (1988). On the evaluation of structural equation models. Journal of the Academy of Marketing Science, 16(1), 74–94.
- Barnett, W.P. (1997). The dynamics of competitive intensity. Administrative Science Quarterly, 42(1), 128–160.
- Barney, J. (1991). Firm resources and sustained competitive advantage. Journal of Management, 17(1), 99–120.
- Bentler, P.M. (1995). EQS 6 structural equations program manual. Encino, CA: Multivariate Software, Inc.
- Bruton, G.D., Ahlstrom, D., & Obloj, K. (2008). Entrepreneurship in emerging economies: Where are we today and where should the research go in the future. *Entrepreneurship: Theory and Practice*, 32(1), 1–14.
- Burgel, O., & Murray, G.C. (2000). The international market entry choices of start-up companies in high-technology industries. *Journal of International Marketing*, 8(2), 33–62.
- Byrne, B.M. (1989). A premier of LISREL: Basic applications and programming for confirmatory factor analytic models. New York: Springer-Verlag.
- Byrne, B.M. (2001). Structural Equation Modeling with AMOS. *Mahwah New Jersey*. London: Lawrence Erlbaum Associates, Publishers.
- Cadogan, J.W., Cui, C.C., & Yeung, E.K. (2003). Export market-oriented behaviour and export performance, the moderating roles of competitive intensity and technological turbulence. *International Marketing Review*, 20(5), 493–513.
- Capron, L., & Hulland, J. (1999). Redeployment of brands, sales forces, and general marketing management expertise following horizontal acquisition: A resourcebased view. Journal of Marketing, 63(2), 41–54.
- Carneiro, J., & Brenes, E.R. (2014). Latin American firms competing in the global economy. Journal of Business Research, 67(5), 831–836.
- Cavusgil, S., Tamer, N., Ghauri, P., & Akcal Akcal, A. (2012). Doing business in emerging markets.
- Cavusgil, S.T., & Knight, G.A. (2015). The born global firm: An entrepreneurial and capabilities perspective on early and rapid internationalization. *Journal of International Business Studies*, 46(1), 3–16.
- Cavusgil, S.T., & Zou, S. (1994). Marketing strategy-performance relationship: An investigation of the empirical link in export market ventures. *Journal of Marketing*, 58(1), 1–21.
- Ciravegna, L, Lopez, L.E., & Kundu, S.K. (2014). Country of origin and network effects on internationalization: A comparative study of SMEs from an emerging and developed economy. *Journal of Business Research*, 67(May), 916–923.
- Classens, S., & Schmukler, S. (2007). "International financial integration through equity markets: Which firms from which countries go global?,". *IMF*.
- Coviello, N.E., & Cox, M.P. (2006). The resource dynamics of international new venture networks. *Journal of International Entrepreneurship*, 4(2/3), 113–132.
- Coviello, N.E., McDougall, P.P., & Oviatt, B.M. (2011). The emergence, advance and future of international entrepreneurship: An introduction to the special forum. *Journal of Business Venturing*, 26(6), 625–631.

- Covin, J.G., & Thomas Lumpkin, G. (2011). Entrepreneurial orientation theory and research: Reflections on a needed construct. *Entrepreneurship: Theory and Practice*, 35(5), 855–872.
- Covin, J.G., & Slevin, D.P. (1989). Strategic management of small firms in hostile and benign environments. *Strategic Management Journal*, 10(1), 75–87.
- Covin, J.G., & Slevin, D.P. (2006). Strategic process effects on the entrepreneurial orientationsales growth rate relationship. *Entrepreneurship Theory and Practice*, 30(1), 57–81.
- Covin, J.G., Slevin, D.P., & Shultz, R.L. (1994). Implementing strategic missions: Effective strategic, structural, and tactical choices. *Journal of Management Studies*, 31(4), 481–503.
- Cui, A.S., Griffith, D.A., & Tamer Cavusgil, S. (2005). The influence of competitive intensity and market dynamism on knowledge management capabilities of multinational corporation subsidiaries. *Journal of International Marketing*, 13(3), 32–53.
- Chandra, Y., Styles, C., & Wilkinson, I. (2009). The recognition of first time international entrepreneurial opportunities. *International Marketing Review*, 26(1), 30–61.
- Day, G.S. (1994). The capabilities of market-driven organizations. *Journal of Marketing*, 58(4), 37–52.
- Day, G. S. (2011). Closing the marketing capabilities gap. Journal of Marketing, 75(October), 183–195.
- Dess, G.G., Tom Lumpkin, G., & Covin, J.G. (1997). Entrepreneurial strategy marketing and firm performance: Tests of contingency and configuration models. *Strategic Management Journal*, 18(9), 677–695.
- Dimitratos, P., Lioukas, S., & Carter, S. (2004). The relationship between entrepreneurship and international performance: The importance of domestic environment. *International Business Review*, 13(1), 19–41.
- Eisenhardt, K.M., & Martin, J.A. (2000). Dynamic capabilities: What are they? Strategic Management Journal, 21(10/11), 1105–1121.
- Fernhaber, S.A., McDougall, P.P., & Oviatt, B.M. (2007). Exploring the role of industry structure in new venture internationalization. *Entrepreneurship: Theory and Practice*, 31(4), 517–542.
- Fleury, A., & Fleury, M.T.L. (2014). Local enablers of business models: The experience of Brazilian multinationals acquiring in North America. *Journal of Business Research*, 67(4), 516–526.
- Fornell, C., & Larcker, D.F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of Marketing Research*, 18(February), 39–50.
- Frechette, M. (2006). Rethinking Latin America: A new approach in U.S. foreign policy. Harvard International Review, 28(2), 28–31.
- Freeman, S., & Tamer Cavusgil, S. (2007). Toward a typology of commitment states among managers of born-global firms: A study of accelerated internationalization. *Journal of International Marketing*, 15(4), 1–40.
- Gabrielsson, M., Gabrielsson, P., & Dimitratos, P. (2014). International entrepreneurial culture and growth of international new ventures. *Management International Review*, 54(4), 445–471.
- George, G.D., Wood, R.D., & Khan, R. (2001). Networking strategy of boards: Implications for small and medium-sized enterprises. *Entrepreneurship and Regional Development*, 13(3), 269–285.
- Gomes, L., & Ramaswamy, K. (1999). An empirical examination of the form of the relationship between multinationality and performance. *Journal of International Business Studies*, 30(1), 173–187.
- Gray, S.R., & Cuevas, T.J. (2005). Regulatory, cognitive and normative factors affecting small business development in northern Mexico. International Journal of Entrepreneurship, 9, 91–100.
- Hair, J.F., Black, W., Babin, B., Anderson, R.E., & Tatham, R.L. (2006). Multivariate data analysis (6th ed.). New Jersey: Pearson-Prentice Hall.
- He, Weijun and Ming Nie (2008), "The impact of innovation and competitive intensity on positional advantage and firm performance " Journal of American Academy of Business 14 (1), 205–10.
- Hewett, K., & Bearden, W.O. (2001). Dependence, trust, and relational behavior on the part of foreign subsidiary marketing operations: Implications for managing global marketing operations. *Journal of Marketing*, 63(October), 51–60.
- Hughes, M., Martin, S. L., Morgan, R. E., & Robson, M. J. (2010). Realizing product-market advantage in high-technology international new ventures: the mediating role of ambidextrous innovation. *Journal of International Marketing*, 18(4), 1–21.
- Hult, G.T., Ketchen, D.J., Griffith, D.A., Chabowski, B.R., Hamman, M.K., Johnson-Dykes, B., ... Tamer Cavusgil, S. (2008). An assessment of the measurement of performance in international business research. *Journal of International Business Studies*, 39, 1064–1080.
- Hult, G., Tomas, M., Snow, C.C., & Kandemir, D. (2003). The role of entrepreneurship in building cultural competitiveness in different organizational types. *Journal of Management*, 29(3), 401–426.
- Hurley, R.F., Tomas, G., & Hult, M. (1998). Innovation, market orientation, and organizational learning: An integration and empirical examination. *Journal of Marketing*, 62, 42–54. IMF (2014). *World economic outlook update*. D.C.: Washington.
- Ireland, R.D., Hitt, M.A., & Sirmon, D.G. (2003). A model of strategic entrepreneurship: The construct and its dimensions. *Journal of Management*, 29(6), 963–989.
- Jantunen, A., Puumalainen, K., Saarenketo, S., & Kylaheiko, K. (2005). Entrepreneurial orientation, dynamic capabilities and international performance. *Journal of International Entrepreneurship*, 3(3), 233–243.
- Jaworski, B.J., & Kohli, A.K. (1993). Market orientation: Antecedents and consequences. Journal of Marketing, 57(3), 53–70.
- Jones, M.V., & Coviello, N.E. (2005). Internationalisation: Conceptualising an entrepreneurial process of behaviour in time. *Journal of International Business Studies*, 36(3), 284–303.
- Jones, M.V., Coviello, N.E., & Tang, Y.K. (2011). International entrepreneurship research (1989–2009): A domain ontology and thematic analysis. *Journal of Business Venturing*, 26, 632–659.

#### S.L. Martin, R.(R.)G. Javalgi / Journal of Business Research xxx (2015) xxx-xxx

- Katsikeas, C.S., Paparoidamis, N.G., & Katsikea, E. (2004). Supply source selection criteria: The impact of supplier performance on distributor performance. *Industrial Marketing Management*, 33, 755–764.
- Keupp, M. M., & Gassmann, O. (2011). Achieving subsidiary integration in international innovation by Managerial 'Tools'. *Management International Review*, 51(2), 213–239.
- Knight, G. (2001). Entrepreneurship and strategy in the international SME. Journal of International Management, 7, 155–171.
- Knight, G. (2000). Entrepreneurship and marketing strategy: The SME under globalization. Journal of International Marketing, 8(2), 12–32.
- Knight, G., & Tamer Cavusgil, S. (2004). Innovation, organizational capabilities, and the born-global firm. *Journal of International Business Studies*, 35, 124–141.
- Kocak, A., & Abimbola, T. (2009). The effects of entrepreneurial marketing on born global performance. *International Marketing Review*, 26(4/5), 439–452.
- Krasnikov, A., & Jayachandran, S. (2008). The relative impact of marketing, research-and-development, and operations capabilities on firm performance. *Journal of Marketing*, 72(4), 1–11.
- Kuivalainen, O., Sundqvist, S., & Servais, P. (2007). Firms' degree of born-globalness, international entrepreneurial orientation and export performance. *Journal of World Business*, 42(3).
- Lederman, D., Messina, J., Pienknagura, S., & Rigolini, J. (2014). Latin American entrepreneurs: many firms but little innovation. Washington, D.C.: The World Bank.
- Lee, C., Lee, K., & Penning, J.M. (2001). Internal capabilities, external networks, and performance: A study on technology-based ventures. *Strategic Management Journal*, 22(6/7), 615–640.
- Lisboa, A., Skarmeas, D., & Lages, C. (2011). Entrepreneurial orientation, exploitative and explorative capabilities, and performance outcomes in export markets: A resourcebased approach. *Industrial Marketing Management*, 40(8), 1274–1284.
- Lopez, LE., Kundu, S.K., & Ciravegna, L. (2009). Born global of born regional? Evidence from an exploratory study in the costa Rican software industry. *Journal of International Business Studies*, 40(7), 1228–1238.
- Lu, J.W., & Beamish, P.W. (2001b). The internationalization and performance of SMEs. Strategic Management Journal, 22(6/7), 565–586.
- Lumpkin, G., & Dess, G. (1996). Clarifying the entrepreneurial orientation construct and linking it to performance. Academy of Management Review, 21(1), 135–172.
- Lumpkin, G. Thomas and Gregory G. Dess (2001), "Linking two dimensions of entrepreneurial orientation to firm performance: The moderating role of environment and industry life cycle.," Journal of Business Venturing, 16 (5), 429–51.
- Makadok, R. (2001). Toward a synthesis of the resource-based and dynamic capability views of rent creation. Strategic Management Journal, 22(5), 387–401.
- Martin, LS., & Javalgi, R. (2013). "The effects of marketing capabilities on performance outcomes in international new ventures: The mediating role of informational knowledge,". *American marketing association marketing summer educators conference*. Boston, MA: American Marketing Association.
- Miller, D. (1983). The correlates of entrepreneurship in three types of firms. Management Science, 29(7), 770–791.
- Miller, D. (2011). Miller (1983) revisited: A reflection on EO research and some suggestions for the future. *Entrepreneurship: Theory and Practice*, 35(5), 873–894.
- Mintzberg, H. (1973). Strategy-making in three modes. California Management Review, 16(2), 44–53.
- Moen, O. (2002). The born globals: A new generation of small European exporters. International Marketing Review, 19(2), 156–175.
- Morgan, N.A., Kaleka, A., & Katsikeas, C.S. (2004). Antecedents of export venture performance: A theoretical model and epirical assessment. *Journal of Marketing*, 68(January), 90–108.
- Morgan, N.A., Katsikeas, C.S., & Vorhies, D.W. (2012). Export marketing strategy implementation, export marketing capabilities, and export venture performance. *Journal* of the Academy of Marketing Science, 40(2), 271–289.
- Morgan, Neil A., Rebecca J. Slotegraff, and Douglas W. Vorhies (2009a), "Linking marketing capabilities with profit growth.," International Journal of Research in Marketing, 26 (4), 284–93.
- Morgan, N.A., Vorhies, D.W., & Mason, C.H. (2009b). Market orientation, marketing capabilities, and firm performance. *Strategic Management Journal*, 30, 909–920.
- Morgan, Neil A., Douglas W. Vorhies, and Bodo B. Schlegelmilch (2006), "Resourceperformance relationships in industrial export ventures: The role of resource inimitability and substituability " Industrial Marketing Management, 35 (5), 621–33.
- Morgan, R.E., & Strong, C.A. (2003). Business performance and dimensions of strategic orientation. *Journal of Business Research*, 56(3), 163–176.
- Mostafa, R., Wheeler, C., & Jones, M.V. (2005). Entrepreneurial orientation, commitment to the internet and export performance in small and medium sized exporting firms. *Journal of International Entrepreneurship*, 3(4), 291–302.
- O'Neill, J., Wilson, D., Purushothaman, R., & Stupnytska, A. (2005). How solid are the BRICs? Goldman Sachs Global Economics Paper. 134.
- Oviatt, B.M., & McDougall, P.P. (1994). Toward a theory of international new ventures. Journal of International Business Studies, 25(1), 45–64.
- Oviatt, B.M., & McDougall, P.P. (2005). Toward a theory of international new ventures. 2004 Decade award winning article,". *Journal of International Business Studies*, 36(1), 29–41.
- Pérez-Batres, L.A., Pisani, M.J., & Doh, J.P. (2010). Latin America's contribution to IB scholarship. Academy of International Busines Insights. 10. (pp. 3–6).
- Ramaswami, S.N., Srivastrava, R.K., & Bhagrava, M. (2009). Market-based capabilities and financial performance of firms: Insights into marketing's contribution to firm value. *Journal of the Academy of Marketing Science*, 37(2), 97–116.
- Ramaswamy, K. (2001). Organizational ownership, competitive intensity, and firm performance: An emprical study of the Indian manufacturing sector. *Strategic Management Journal*, 22(10), 989–998.

- Rasmussan, E., Madsen, T.K., & Evangelista, F. (2001). The founding of the born global company in Denmark and Australia: Sensemaking and networking. Asia Pacific Journal of Marketing and Logistics, 13(3), 75–107.
- Rauch, A., Wiklund, J., Frese, M., & Lumpkin, G. (2004). Entrepreneurial orientation and business performance: Cumulative empirical evidence. In W.D. Bygrave, C.G. Brush, M.L.P. Davidson, G.D. Meyer, J. Fiet, J. Sohl, A. Green, A. Zacharakis, & R.T. Harrison (Eds.). Frontiers of entrepreneurship research. Babson College: Welleslev. MA.
- Rialp, A., Rialp, J., & Knight, G.A. (2005). The phenomenon of early internationalizing firms: What do we know after a decade (1993–2003) of scientific inquiry? *International Business Review*, 14, 147–166.
- Ripollés, M., & Blesa, A. (2012). International new ventures as "small multinationals": The importance of marketing capabilities. *Journal of World Business*, 47(2), 277–287.
- Ripollés, M., Menguzzato-Boulard, M., & Sanchez-Peinado, L. (2007). Entrepreneurial orientation and international commitment. *Journal of International Entrepreneurship*, 5(3–4), 65–83.
- Robson, M.J., Katsikeas, C.S., & Bello, D.C. (2008). Drivers and performance outcomes of trust in international strategic alliances: The role of organizational complexity. *Organization Science*, 19(4), 647–665.
- Salvo, A. (2007). Cemex: Making global markets. In D. Spulberg (Ed.), *Global competitive strategy*. Cambridge, New York, Melbourne, Madrid, Cape Town, Singapore, Sao Paulo: Cambridge University Press.
- Sapienza, H.J., Autio, E., George, G., & Zahara, S. (2006). A capabilities perspective on the effects of early internationalization on firm survival and growth. Academy of Management Review, 31(4), 914–933.
- Shrader, R.C. (2001). Collaboration and performance in foreign markets: The case of young high-technology manufacturing firms". Academy of Management Journal, 1, 45–60.
- Shrader, R.C., Oviatt, B.M., & McDougall, P.P. (2000). How new ventures exploit trade-offs among international risk factors: Lessons for the accelerated internationalization of the 21st century. Academy of Management Journal, 43(6), 1227–1247.
- Slater, S.F., & Narver, J.C. (1994). Does competitive environment moderrate the market orientation performance relationship? *Journal of Marketing*, 58(1), 46–55.
- Slevin, D.P., & Terjesen, S. (2011). Entrepreneurial orientation: Reviewing three papers and implications for further theoretical and methodological development. *Entrepreneurship: Theory and Practice*, 35(5), 973–987.
- Smart, D.T., & Conant, J.S. (1994). Entrepreneurial orientation, distinctive marketing competencies, and organizational performance. *Journal of Applied Business Research*, 10(3), 28–38.
- Stam, W., & Elfring, T. (2008). Entrepreneurial orientation and new venture performance: The moderating role of intra- and extra-indsutry social capital. Academy of Management Journal, 51(1), 97–111.
- Teece, D., Pisano, G., & Shuen, A. (1997). Dynamic capabilities and strategic management.
- Strategic Management Journal, 18(7), 509–533.
- UNCTAD (2014). World Investment Report. UnitedNations. New York and Geneva. Vorhies, D.W., & Morgan, N.A. (2005). Benchmarking marketing capabilities for sustained
- competitive advantage. Journal of Marketing, 69(1), 80–94.
- Vorhies, D.W., & Morgan, N.A. (2003). A configuration theory assessment of marketing organization fit with business strategy and its relationship with marketing performance. *Journal of Marketing*, 67(January), 100–115.
- Walker, B.J., Kirchmann, W., & Conant, J.S. (1987). A method to improve response to industrial mail surveys. Industrial Marketing Management, 16(4), 305–314.
- Walker, O.C., & Ruekert, R.W. (1987). Marketing's role in the implementation of business strategies: A critical review and conceptual framework. *Journal of Marketing*, 51(July), 15–33.
- Wang, C.L. (2008). Entrepreneurial orientation, learning orientation, and firm performance. Entrepreneurship: Theory and Practice, 32(4), 635–657.
- Weerawardena, J., Liesch, P.W., Mort, G.S., & Knight, G.A. (2007). Conceptualizing accelerated internationalization in the born global firm: A dynamic capabilities perspective. *Journal of World Business*, 42(3), 294–306.
- Wiklund, J. (1999). The sustainability of the entrepreneurial orientation–performance relationship. *Entrepreneurship: Theory and Practice*, 24(1), 37–48.
- Wiklund, J., & Shepherd, D. (2003). Knowledge-based resources, entrepreneurial orientation, and the performance of small and medium-sized business. *Strategic Management Journal*, 24(13), 1307–1314.
- Wiklund, J., & Shepherd, D. (2005). Entrepreneurial orientation and small business performance: A configuration approach. *Journal of Business Venturing*, 20(1), 71–91.
- Wiles, M.A., Morgan, N.A., & Rego, LL (2012). The effect of brand acquisition and disposal on stock returns. *Journal of Marketing*, 76(1), 38–58.
- Wood, E., Khavul, S., Pérez-Nordtvedt, L., Prakhya, S., Dabrowski, R.V., & Zheng, C. (2011). Strategic commitment and timing of internationalization from emerging markets: Evidence from China, India, Mexico and South Africa. *Journal of Small Business Management*, 49(2), 252–282.
- World Bank (2014). World development indicators. DC: Washington.
- Zahra, SA. (1991). Predictors and financial outcomes of corporate entrepreneurship: An explorative study. *Journal of Business Venturing*, 6, 259–285.
- Zahra, S.A. (2006). Entrepreneurship and dynamic capabilities: A review, model and research agenda. *Journal of Management Studies*, 43(4), 917–955.
- Zahra, S.A., & Covin, J.G. (1995). Contextual influences on the corporate entrepreneurshipperformance relationship: A longitudinal analysis. *Journal of Business Venturing*, 10, 43–58.
- Zahra, S.A., & Das, S.R. (1993). Innovation strategy and financial performance in manufacturing companies: An empirical analysis. *Production and Operations Management*, 2(1), 15–37.

12

# **ARTICLE IN PRESS**

S.L. Martin, R.(R.)G. Javalgi / Journal of Business Research xxx (2015) xxx-xxx

- Zahra, S.A., & Garvis, D.M. (2000). International corporate entrepreneurship and firm performance: The moderating effect of international environment hostility. *Journal of Business Venturing*, 15(5/6), 469–492.
- Zahra, S.A., Duane Ireland, R., & Hitt, M.A. (2000). International expansion by new venture firms: International diversity, mode of market entry, technological learning, and performance. Academy of Management Journal, 43(5), 925–950.
- Zhou, L., Wu, A., & Barnes, B. (2012). The effects of early internationalization on performance outcomes in young international ventures: The mediating role of marketing capabilities. *Journal of International Marketing*, 20(4), 25–45.
- capabilities. Journal of International Marketing, 20(4), 25–45.
  Zou, S., Fang, E., & Zhao, S. (2003). The effect of export marketing capabilities on export performance: An investigation of Chinese exporters. Journal of International Marketing, 11(4), 32–55.